Author's response to reviews

Title: Congenital Rubella Syndrome in Iran

Authors:
Jila Sadighi (jilasadigh@yahoo.com)
Hassan Eftekhar (info@ihsr.ac.ir)
Kazem Mohammad (info@ihsr.ac.ir)

Version: 3 Date: 20 December 2004

Author's response to reviews:

Dear editor

The changes in my manuscript are:

1- English improved
2- Formula is describe in two lines (in the last version was in one line)
3- The arrangement of references is changed (14 is excluded and 12 is moved to 15)
4- Used the children 1-4 years instead of birth rate in formula (therefore, numbers in formula are changed)
5- Summarized some parts of discussion

Best Regards,

Dr. Jila Sadighi